Author's response to reviews

Title: Triple malignancy in a single patient including a cervical carcinoma, a basal cell carcinoma of the skin and a neuroendocrine carcinoma from an unknown primary site: a case report and comprehensive review of the literature.

Authors:

Mohamed Mesmoudi (mohamedmesmoudi@yahoo.fr)
Saber Boutayeb (boutayebdr@yahoo.fr)
Tarik Mahfoud (mahfoudtarik@hotmail.fr)
Rachid Aasab (rachid30-@hotmail.com)
Nabil Ismaili (ismailinabil@yahoo.fr)
Meryem Glaoui (m_glaoui@hotmail.com)
Hassan Errihani (h_errihani@yahoo.fr)

Version: 4 Date: 5 August 2011

Author's response to reviews:

August 5th, 2011
Subject: Submission of the revised version of the manuscript.
Dear Editor,
Please find attached a revised version of the manuscript. Dates were removed and replaced as requested.
I hope this version will be the definitive one.
Sincerely,
Mohamed Mesmoudi.

Department of medical oncology, National institute of oncology, Rabat, Morocco.
mohamedmesmoudi@yahoo.fr